Author’s response to reviews

Title: Comparing multidisciplinary and brief intervention in employees with different job relations on sick leave due to low back pain: protocol of a randomised controlled trial

Authors:

Pernille Pedersen (Pernille.Pedersen@stab.rm.dk)
Claus Nielsen (claus.vinther@rm.dk)
Morten Andersen (Morten.Andersen@midt.rm.dk)
Vivian Langagergaard (vivian.langagergaard@vest.rm.dk)
Anders Boes (andeboes@rm.dk)
Ole Jensen (olejesen@rm.dk)
Chris Jensen (chris.jensen@arbeidoghelse.no)
Merete Labriola (Merete.Labriola@stab.rm.dk)

Version: 1 Date: 13 Apr 2017

Author’s response to reviews:

Dear editor,

We appreciate your comments for the manuscript "Comparing multidisciplinary and brief intervention in employees with different job relations on sick leave due to low back pain: protocol of a randomised controlled trial" (PUBH-D-17-00663).

The study has been approved by the Central Denmark Region Committees on Health Research Ethics. We have forwarded the approval for the study by mail to BMCSeriesEditorial@biomedcentral.com. The study received no specific grant from any funding agency in the public, commercial, or not-for-profit sectors, but was financed by the Regional Hospital Silkeborg and DEFACTUM, Central Denmark Region. Therefore, we can not forward any funding approvals.
We are currently collecting data for the study and have not published or submitted any papers on the results of the study yet.

The Declarations section has been adjusted according to the guidelines. We have uploaded a new version of the manuscript, which we hope you will consider for publication in BMC Public Health.

On behalf of the authors,

Pernille Pedersen